Author's response to reviews

Title: An electronic portfolio for quantitative assessment of surgical skills in undergraduate medical education

Authors:

Serafin Sanchez Gomez (sanchezsg@ono.com)
Elisa Maria Cabot Ostos (emcabot@us.es)
Juan Manuel Maza Solano (juan.maza.solano@gmail.com)
Tomas Francisco Herrero Salado (tomasherrrero@gmail.com)

Version: 6 Date: 17 February 2013

Author's response to reviews: see over
To the Editor of BMC Medical Education

Seville, February 16th 2013

Dear Editor:

The authors are deeply grateful again to the comments issued by the editor. Following your advice, we have submitted the manuscript MS: 1057603099656817 - An electronic portfolio for quantitative assessment of surgical skills in undergraduate medical education to Edanz (English Editing for scientists) for wide and substantial copyediting.

Editing process has finished and now we are greatly satisfied with Edanz job. We hope that the extensive correction of language and grammar of the paper would satisfy your requirements.

All authors and coauthors worked in the revision of the last version of the manuscript and have given their final approval to be published if it is accepted.

Yours sincerely,

[Signature]

Corresponding author: Prof. Dr. Serafín Sánchez Gómez